

Received 29 November 2022, accepted 30 December 2022, date of publication 11 January 2023, date of current version 6 February 2023. Digital Object Identifier 10.1109/ACCESS.2023.3236084

TOPICAL REVIEW

The Role of Artificial Intelligence in Future Rehabilitation Services: A Systematic Literature Review

CIRO MENNELLA[®], UMBERTO MANISCALCO[®], GIUSEPPE DE PIETRO, AND MASSIMO ESPOSITO[®]

Institute for High-Performance Computing and Networking (ICAR), Research National Council of Italy (CNR), 87036 Rende, Italy Corresponding author: Umberto Maniscalco (umberto.maniscalco@icar.cnr.it)

ABSTRACT Artificial intelligence technologies are considered crucial in supporting a decentralized model of care in which therapeutic interventions are provided from a distance. In the last years, various approaches have been proposed to support remote monitoring and smart assistance in rehabilitation services. Comprehensive state-of-the-art of machine learning methods and applications is presented in this review. Following PRISMA guidelines, a systematic literature search strategy was led in PubMed, Scopus, and IEEE Xplore databases. The search yielded 519 records, resulting in 35 articles included in this study. Supervised and unsupervised machine learning algorithms were identified. Unobtrusive capture motion technologies have been identified as strategic applications to support remote and smart monitoring. The main tasks addressed by algorithms were activity recognition, movement classification, and clinical status prediction. Some authors evidenced drawbacks concerning the low generalizability of the results retrieved. Artificial intelligence-based applications are likely to impact the delivery of decentralized rehabilitation services by providing broad access to sustained and high-quality therapy. Future efforts are needed to validate artificial intelligence technologies in specific clinical populations and evaluate results reliability in remote conditions and home-based settings.

INDEX TERMS Digital therapeutics, e-health, remote monitoring, intelligent systems, deep learning, machine learning.

I. INTRODUCTION

The current model of care is resource-demanding and already faces major challenges in coping with an increasing number of patients due to demographic change, the paucity of skilled healthcare professionals and economic pressure to minimize healthcare costs [1].

Traditional rehabilitation services strongly rely on physical therapy sessions, which are based on one-to-one interactions with healthcare practitioners during a hospital stay (above all during the acute and sub-acute phase of the disease) or as part of periodic visits to specialized clinics (mostly during the chronic phase). Despite the clinical-centered model allowing healthcare professionals to closely monitor and

The associate editor coordinating the review of this manuscript and approving it for publication was Santosh Kumar^(D).

support patients and their families physically and emotionally, it presents some limitations. A clinical-centered model that heavily relies on physical access to medical facilities and prolonged interactions with trained specialists makes events such as the coronavirus pandemic (COVID-19) and the increased demand for assistance to chronic patients challenging to manage [2], [3]. Distancing measures necessary during a pandemic phase and the increasing management issues avoiding the overload of the healthcare system could lead to a reduction in rehabilitation visits. Moreover, the lack of human resources could affect the dose and quality of therapy patients receive, limiting rehabilitation to relatively short periods that might be insufficient to achieve functional recovery [4], [5].

Despite the growing evidence that intensive high-dose rehabilitation positively impacts functional recovery even long after the disease, therapy dose is typically rather low at all stages of the continuum of care. In fact, after clinic discharge, there is limited support for ongoing rehabilitation management at home, which is necessary for the maintenance of functional outcomes in physical activities [6], [7]. To this end, there is a need for a sustainable approach that facilitates patients' long-term management more efficiently [8], [9], [10]. Many efforts are needed to shift care services toward a decentralized model opposite the conventional therapy approach.

New developments are emerging as possible answers to rehabilitation service limitations related to time, distance, difficult terrains, costs, and limited access to clinical facilities [11]. Some proposed solutions span from a live communication stream in which rehabilitation professionals guide and encourage patients while monitoring their progress to more advanced solutions involving various sensor and information and communication technologies (ICTs), sometimes coupled with artificial intelligence (AI) algorithms to support the monitoring and assistance in remote contexts [12], [13].

AI applications are considered to play a key role in further establishing and supporting a decentralized rehabilitation model in which intelligent connected tools will be employed to assist clinical decision-making, and health outcomes monitoring [14], [15]. Many AI-based methods and solutions have been proposed in recent years to support the future challenge of enabling assisted physical therapy and assessments in a minimally supervised and decentralized manner, ideally at the patient's home. However, to the best of the authors' knowledge, no published works provided a comprehensive review of machine learning methods and applications used for remote monitoring and assistance in the rehabilitation context. Some existing works in literature have provided an overview of the role of machine learning algorithms combined with specific technologies used for rehabilitation issues, such as wearable sensors [16], [17] and vision-based motion capture technologies [18]. More specifically, a recent work emphasized machine learning methods for movement evaluation in rehabilitation programs using motion capture systems [19]. Other works highlighted the progress in machine learning for automated evaluation of patient performance and recovery, introducing data analytics to improve the effectiveness and efficiency of care for physical rehabilitation [20].

Due to the growing importance of AI-based applications to support remote rehabilitation procedures, with the present work, for the first time, the authors aim to provide succinctly state-of-the-art about AI-based machine learning solutions supporting the promotion and delivery of physical therapy from a distance. This paper is targeted at researchers and technical developers in the field of intelligent systems for smart monitoring and assessment in rehabilitation. The goal is to highlight key aspects where emerging AI-based machine learning approaches may help solve peculiar clinical practice issues. Both advantages and drawbacks of existing approaches and future implications of AI-based solutions supporting decentralized rehabilitation services are pertinently discussed by the authors. The review and recommendations provided in this paper aim to guide the design of the next generation of AI applications in rehabilitation.

The paper is organized as follows. Section II describes the review methods and article selection procedure. Section III presents the results achieved and surveys of machine learning methods and applications proposed in recent years. Section IV provides a discussion about the potential direction of future research avenues. Section V concludes the work.

II. MATERIALS & METHODS

A. STUDY DESIGN AND CONTRIBUTION

This work aims to provide general and comprehensive content with the scope to better understand the recent field development.

Considering the aim of this study, a systematic literature review is an appropriate research design and strategy accordingly to Snyder guidelines [21]. Therefore, the results are synthesized and discussed according to a qualitative research approach that entails undertaking a thorough critique of each piece of text and identifying recurring themes from different works, which finally contribute to forming the basis for the critical conclusions of analyzed studies [22].

To the authors' knowledge, this article would be the first review to discuss the challenges, opportunities, and solutions of AI applications for rehabilitation services provided in remote and decentralized conditions. Most of the current literature addressing AI technologies and methods in rehabilitation is focused on specific applications or a specific area of research. This work offers a broader overview of different research areas and applications. Table 1 highlights the major differences between this review article with other articles published in the rehabilitation field. In addition, this work highlights the richness and complexity of each application domain to guide future research and developments.

The main contribution of the article is outlined as follows:

- AI-based systems and technologies for monitoring and assistance in rehabilitation are identified and discussed along with their practical implications;
- machine learning algorithms that find their application in rehabilitation are classified and discussed according to the state-of-art literature;
- machine learning methods and approaches are outlined and discussed according to the clinical application scopes from a practical point of view;
- challenges and future trends of AI applications for rehabilitation services provided at a distance are outlined.

B. LITERATURE SEARCH STRATEGY

A literature search was conducted in PubMed, IEEE Xplore, and Scopus databases by covering the period from 2010 to May 2022. A structured search strategy was performed in each electronic database concerning AI-based machine learning methods and applications supporting remote monitoring and assistance in the rehabilitation context. The same search

TABLE 1. Key differences between this review and existing literature.

Paper	Key differences		
[17]	The objective of the study was to survey ex-		
	ercise evaluation in post-stroke rehabilitation		
	by means of wearable devices and machine		
	learning algorithms.		
[18]	The study reviewed the evolution of RGB-		
	D sensors for musculoskeletal health monitor-		
	ing, also evidencing computational algorithms		
	based on machine learning.		
[19]	The study reviewed computational approaches		
	for evaluating patient performance in rehabili-		
	tation programs using motion capture systems.		
[20]	The study provided a non-systematic brief		
	overview of big data analysis to optimize out-		
	patient rehabilitation.		

TABLE 2. Electronic search strategy.

Database	Query
PubMed	("artificial intelligence" OR "deep learn-
	ing" OR "machine learning") AND (
	(tele OR remote OR home) AND
	(evaluation OR monitoring OR supervi-
	sion OR assessment OR interaction OR
	feedback OR communication)) AND
	("physical therapy" OR exercise OR
	"physical rehabilitation")
IEEE Xplore	(artificial intelligence OR deep learning
	OR machine learning) AND (tele OR
	remote OR home) AND (evaluation OR
	monitoring OR supervision OR assess-
	ment OR interaction OR feedback OR
	communication) AND (physical therapy
	OR exercise OR physical rehabilitation)
Scopus	TITLE-ABS-KEY (("artificial intelli-
	gence" OR "deep learning" OR "ma-
	chine learning") AND ((tele OR
	remote OR home) AND (evaluation
	OR monitoring OR supervision OR as-
	sessment OR interaction OR feedback
	OR communication)) AND ("physi-
	cal therapy" OR exercise OR "physical
	rehabilitation")) AND (LIMIT-TO (
	LANGUAGE, "English"))

string was used for PubMed, IEEE Xplore, and Scopus, with the only difference due to the syntax required by the database. Performed queries are shown in Table 2.

C. STUDY SELECTION PROCESS

This review was carried out according to the Cochrane Collaboration methodology [23]. The article selection process





FIGURE 1. PRISMA flowchart of the results from the literature search.

based on the PRISMA guidelines [24] is schematically presented in Figure 1.

After removing duplicated articles, titles and abstracts were explored, and then full-text screening was conducted according to inclusion and exclusion criteria for eligibility. Only the most recent study was retained when the same authors published several studies on the same research initiative.

Inclusion criteria were: (i) articles describing AI methods and applications; (ii) articles concerning both clinical and technical aspects of solving issues regarding decentralized services in rehabilitation; (iii) articles that must be written in English; (iv) articles published after 2010 (considering the recent employment of AI applications in healthcare field it would not be necessary to review older articles). Exclusion criteria were: (i) machine learning techniques not addressed by authors; (ii) technical laboratory studies describing only systems developments not covering clinical issues; (iii) only the most recent work was retained when advancements from previous research initiatives of the same author were retrieved; (v) review articles.

D. DATA EXTRACTION PROCESS

Data extraction was performed manually. The extracted data included: (i) characteristics of the system involved in acquiring data; (ii) main features concerning the machine learning algorithms and results retrieved; (iii) information about the clinical application field evidencing rehabilitation scopes and therapy issues; (iv) the study aims and its characteristics evidencing drawbacks and advantages reported by respective authors. The "Scimago Journal & Country Rank" (SJR) database was used to extract quartile metrics to assess the quality of selected studies.

III. RESULTS

A. DATA SYNTHESIS

The initial paper search yielded a total of 519 results. After the preliminary screening, 93 articles were selected for eligibility. Applying the inclusion and exclusion criteria led to 35 articles related to AI-based machine learning solutions supporting the monitoring and assistance of physical therapy from a distance. Figure 1 schematically presents the results of each step of the review process.

The data extracted from reviewed articles are summarized and presented as follows: distribution of the number of articles according to the SJR quality quartile metrics over the years in Figure 2, characteristics of systems and machine learning methods in Table 3, information concerning the clinical application context in Table 4, implications and drawbacks of the selected studies in Table 5.



FIGURE 2. Distribution of the number of articles published every year and quality quartile ranking according to SJR database. Abbreviations: N.R. = not reported.

Wearable capture motion systems resulted in applications most used in selected studies. Almost all the reviewed articles addressed classification, and regression algorithms, except in work [25], which applied an unsupervised learning technique for a clustering task. Figure 3 graphically summarizes all the results retrieved.

Most AI-based machine learning algorithms have been trained to recognize activities and evaluate specific movements remotely, predicting motion data from more complex systems using unobtrusive and low-cost technology. Minor studies focused on the prediction of the patient's clinical status.

Most of the included works resulted in feasibility and validation studies conducted in supervised laboratory settings involving healthy subjects. Figure 4 compares articles for applications scope and SJR quality quartile metrics to evidence some existing literature shortcomings.



FIGURE 3. Results synthesis: A) capture motion technologies, B) machine learning tasks, C) machine learning applications.



FIGURE 4. Radar chart comparing the reviewed articles for SJR quartile quality metrics and their application scopes. Abbreviations: N.R. = not reported.

B. AI-BASED SYSTEMS AND TECHNOLOGIES

The remote delivery of rehabilitation services should enable therapists to optimize the timing, intensity, and duration of therapy which is often not possible due to the constraints of face-to-face treatment protocols of the traditional healthcare model. Therefore, ICTs used in decentralized rehabilitation services allow facing traditional logistic problems making treatments access equitable to geographically remote individuals.

Applications to date used to support decentralized services can be categorized as: i) synchronous systems using real-time electronic communication via simple webinar platforms through which healthcare professionals provide audio and video support for the patient, and ii) asynchronous systems using store-and-forward communication involving data and collection through distributed devices [60], [61]. Healthcare professionals typically use synchronous systems to interact with patients on a live communication stream, guiding and encouraging patients by monitoring their progress. Asynchronous approaches, in general, involve the acquisition, transmission, and elaboration of data to generate an appropriate action in patient care with associated decision support [13]. Such approaches have attracted recent researchers' interest, as evidenced by the results of reviewed articles. Some attempts have been made to provide real-time audio and video suggestions with asynchronous technologies.

TABLE 3. Summary of the paper lists and information related methods, techniques, and results retrieved. Abbreviations: ML: machine learning, LSTM: long short-term memory, DT: decision tree, MAE: mean absolute error, IMU: inertial measurement unit, KNN: K-nearest neighbor, SVM: support vector machine, RF: random forest, MLP: multilayer perceptron, CNN: convolution neural network, SOM: self-organized map, CART: classification regression tree, GCN: graph convolutional network, NB: naive bayesian, HMM: hidden Markov model, ELM: extreme learning machine.

Reference (Year)	System	ML Method	Aim	Results
[26](2022)	Kinect-based	LSTM, DT XGBoost,	Estimation of weight-	LSTM resulted the best
	exergame platform	Linear Regression	shifting during	estimation model
			exergaming using	
[27](2022)	Virtual reality	DT gradient boosted	Predict joint angles and	Accurate predictions:
[27](2022)	exergame platform	DI gradient boosted	torques with motion	ioint angles (MAE <
			tracking data	0.78°), joint torques
				(MAE < 2.34 Nm)
[28](2022)	Smart textile garment	KNN	Activity classification	Accuracy of 0.94
	composed by optical		for smart monitoring	
F001(0000)	fiber sensors and IMU			
[29](2022)	Virtual coach applica-	SVM, DT, RF, KNN,	Evaluate the	SVM resulted in
	tion based on INIU sen-	MLP	performance of six	the most accurate
	5015		contexts: i) recognition	of 0.88 and 0.91
			and evaluation in a	in recognition and
			single classifier, ii)	evaluation task for
			recognition of correct	a single classifier;
			exercises, iii) a two-	accuracy of 0.96
			stage approach that	in recognition and
			first recognizes the	between 0.93-1.00 in
			evaluates it	the two-stage approach
[30](2022)	Motion capture system	CNN-LSTM	Estimating lower limb	The estimated results
LJ(/	based on computer vi-		joint angular velocities	show the highest cor-
	sion			relation 0.93 in squat
				and 0.92 in walking on
				treadmill action
[31](2021)	Accelerometer-based	SVM, RF, MLP	Detect and track reha-	MLP model accuracy >
[32](2021)	IMI L-based	CNN	Measure the rate	0.80 For the binary
[52](2021)	smartwatch	CIVIN	and patterns of	classification task.
			participation in	the CNN model
			rehabilitation sessions,	achieved an accuracy
			assess the dose-	of 0.95, a sensitivity of
			response between	0.94; a specificity of
			physiotherapy activity	0.97; and AUC of 0.99 .
			and recovery, and	ror the multi-class
			factors predictive	classifier achieved an
			of participation	accuracy of 0.90 and
				an F1-score of 0.82
[25](2021)	Exergame platform	SOM, CART	Develop a new	Four clusters were
	based on Microsoft		method to personalize	selected by SOM
	Kinect motion sensor		exergame setting	with high prediction
	camera		(range of motion,	accuracy in temale
			moving time points	$R^{2}=0.93$ MAF=1.30
			repetitions)	and male group
			x · · · · · · · · · · · · · · · · · · ·	(RMSE=3.75,
				R2=0.99, MAE=3.03);
				CART prediction
				obtained high accuracy $(B_{2}-0.80)$
				(R2=0.89)

TABLE 3. (Continued.) Summary of the paper lists and information related methods, techniques, and results retrieved. Abbreviations: ML: machine learning, LSTM: long short-term memory, DT: decision tree, MAE: mean absolute error, IMU: inertial measurement unit, KNN: K-nearest neighbor, SVM: support vector machine, RF: random forest, MLP: multilayer perceptron, CNN: convolution neural network, SOM: self-organized map, CART: classification regression tree, GCN: graph convolutional network, NB: naive bayesian, HMM: hidden Markov model, ELM: extreme learning machine.

[33](2021)	Motion Coach app based on computer vision pose estimation technique	CNN	Evaluate the ability of the system to detect and correct exercises form	No significant differences between: i) Motion Coach app and physiotherapists=0.82 inter-rater agreements; ii) physiotherapist 1 and physiotherapist 2=0.83 inter-rater agreements
[34](2021)	Data glove composed by flex sensors, force- resistive sensors and IMU	DT XGBoost, Logistic Regression	Automated assessment in-home rehabilitation session	The Logistic Regres- sion classifier outper- formed the XGB clas- sifier with an aver- age accuracy of 0.85 among all tasks
[35](2021)	Motion capture system based on computer vi- sion	GCN	Assessing exercise per- formance	GCNmethodsachievedbetterperformancethanCNNandLSTM(MAE=0.02)
[36](2021)	Motion capture system based on computer vi- sion	DT, RF, NB, KNN, SVM	Automatic evaluation of home-based rehabilitation exercise	SVM achieved the best accuracy 0.98
[37](2021)	Wearable-based motion capture system	SVM	Classifying single-axis spinal motion using data streams from stretch sensors for exercise recognition	SVM achieved an ac- curacy of 0.85
[38](2020)	Exergame platform composed by electromyography and inertial sensors	SVM, RF	Develop real-time gesture recognition for feedback and control during exergame session	SVM classifier showed good performance (F1- score of 0.82-0.90)
[39](2020)	Microsoft Kinect cam- era coupled with Wii- Balance Board	CNN, RF	Develop a balance evaluation system using inexpensive and portable sensors to measure the subject's balance level	CNN reduced the estimation error by about 10% compared with other state-of- the-art models for balance evaluation; the RF achieved high sensitivity and specificity to classify the four levels of balance problems (>0.80) and the two categories fall risk /no fall risk (>0.90)
[40](2020)	IMU-based smartwatch and mobile app	CNN	Recognize the type and frequency of rehabil- itation exercises per- formed	The CNN with personal data involving accelerometer combined with gyroscope data was the most accurate (0.99)

TABLE 3. (Continued.) Summary of the paper lists and information related methods, techniques, and results retrieved. Abbreviations: ML: machine learning, LSTM: long short-term memory, DT: decision tree, MAE: mean absolute error, IMU: inertial measurement unit, KNN: K-nearest neighbor, SVM: support vector machine, RF: random forest, MLP: multilayer perceptron, CNN: convolution neural network, SOM: self-organized map, CART: classification regression tree, GCN: graph convolutional network, NB: naive bayesian, HMM: hidden Markov model, ELM: extreme learning machine.

[41](2020)	IMU-based wearable sensor	CNN, SVM, RF, KNN, MLP	Automatic remote ex- ercise recognition and repetition counting	CNN achieved higher accuracy for exercise recognition (0.97) and for repetition counting task (0.90)
[42](2020)	Exergame platform based on pose estimation method	CNN	Comparing a transfer learning-based pose es- timation model to the Kinect system and a gold-standard 3DMo- Cap system	No statistical differ- ence between CNN- based pose estimation, Kinect and the gold standard system
[43](2020)	IMU-based wearable sensor	CNN	Segmentation of exer- cise repetitions	Flexion and abduction exercises achieved high accuracy (0.90- 0.95), scapular movement poor accuracy (<0.85)
[44](2019)	Socially assistive robot	CNN, LSTM	Exercise recognition	CNN-LSTM achieved accuracy of 0.99
[45](2019)	IMU-based wearable sensor	DT, SVM, KNN, RF	Recognition of shoul- der rehabilitation activ- ities	Optimal predictive accuracies for SVM (0.97) and RF (0.80)
[46](2019)	Conversational agent- assisted health coach- ing system	SVM	Provide personalize suggestions to support and engage health intervention	After 1-month individ- uals showed generally positive reaction to a virtual agent providing them useful health in- terventions and feed- backs
[47](2019)	Kinect-based digital coach	SVM, RF	Personalized and auto- matic remote monitor- ing and evaluation	Accuracy of SVM in patient error identifica- tion above 0.90; accu- racy of RFs model for task recommendations ranged between 0.81- 0.91 in all classes
[48](2018)	Kinect-based platform	НММ	Automatically detects the correctness of the executed movement to provide real-time feed- back to the patient	High accuracy in the evaluation of the move- ments (0.92 of correct classification)
[49](2018)	Exergame platform composed by Kinect, leap motion and wearable wrist-band	Logistic Regression	Discriminating between good or bad activity performance during various serious games	High classification ac- curacy for all exercise evaluated
[50](2018)	IMUs-based wearable sensor	Logistic regression, KNN, RF	Detecting goal- directed movements and assessing the quality of motor performance	Correct movements were classified with a predictive accuracy of 0.87, the incorrectly performed exercise movements with an accuracy of 0.84

TABLE 3. (Continued.) Summary of the paper lists and information related methods, techniques, and results retrieved. Abbreviations: ML: machine learning, LSTM: long short-term memory, DT: decision tree, MAE: mean absolute error, IMU: inertial measurement unit, KNN: K-nearest neighbor, SVM: support vector machine, RF: random forest, MLP: multilayer perceptron, CNN: convolution neural network, SOM: self-organized map, CART: classification regression tree, GCN: graph convolutional network, NB: naive bayesian, HMM: hidden Markov model, ELM: extreme learning machine.

[51](2018)	Kinect	Linear Regression	Solve self-occlusion errors in motion analysis mapping the noisy Kinect- based measurements to a more accurate benchmark system measurement	The denoising method resulted in significantly better than a bench- mark system
[52](2018)	smartwatch	KINN, RF, SVM, CINN	Exercise recognition	civic accuracy ranged be- tween 0.88-0.99
[53](2018)	Wearable-based motion capture system	Linear regression, RF, MLP	Estimate the knee joint angle based on the strain sensor data	RF model achieved best results (intra- subject evaluation: walking task MAE=1.94, R2=0.97; flexion exercises MAE=3.02, R2=0.97; inter-subject evaluation: walking task MAE=4.14, R2=0.90; flexion exercises MAE=6.97, R2=0.90)
[54](2018)	IMU-based motion capture system	Logistic regression, SVM, DT AdaBoost, RF	Classify correct and in- correct exercise repeti- tions	SVM and RF achieved the best performance accuracy (0.89-0.97)
[55] (2017)	Leap motion system	SVM, KNN	Develop a hand ges- ture recognition algo- rithm to monitor re- mote rehabilitation ex- ercise	SVM and KNN achieved an accuracy of 0.97
[56] (2017)	Multisensor wearable system	Softmax regression, SVM, RF, MLP	Activity recognition	RF achieved highest accuracy 0.99
[57] (2015)	Smartphone-based Tele-Rehabilitation System	Linear regression	Estimating the shoul- der joint ROM by us- ing only data from one accelerometer sensor.	Linear regression achieved an accuracy of R2=0.99
[58] (2012)	Accelerometer-based motion capture system	ELM	Develop Brunnstrom stage automatic evaluation	ELM achieved an ac- curacy of 0.92
[59] (2012)	IMU-based wearable motion capture systems	DT AdaBoost	Recognize patients' er- ror during prescribed exercise	DT detect classes in multi-label data with 0.75 sensitivity, 0.90 specificity and 0.80 ac- curacy

Retrieved systems have been used to guide and assist motion while collecting objective data on movement quality and encouraging remote rehabilitation sessions *via* digital therapeutic approaches, providing access to quality therapy from a distance without needing constant supervision by a therapist on site. Different digital strategies have been individualized: exergame solutions that engage the patient to perform some functional movements providing relevant

Reference	Rehabilitation Field	Therapy	Subjects
[26]	Balance	A custom exergame for balance	Healthy older subjects (N=12)
		training that elicits medio-lateral	
		weight shifts from the user	
[27]	Shoulder injuries	Shoulder exercise: rotation, side	Healthy subjects (N=5)
	~	arm raise, forward arm raise, ex-	
		ternal rotation, abducted rotation.	
		press, circles	
[28]	Activity promotion	Daily activities: standing, sitting,	Healthy subjects (N=4)
[[-0]	From the promotion	squatting, up-and-down arms,	
		walking and running	
[29]	Activity promotion in elderly	A set of eight exercises focused	Healthy subjects (N=30)
[=>]		on the lower or upper limbs: knee	
		flex-extension squats and hip ab-	
		duction, natural gait, and heel-tip	
		toe gait, squeezing, elbow flex-	
		extension and extension of arms	
		overhead	
[30]	Physical activity	Squat, walking on treadmill	Healthy subject(N=33)
[31]	Stroke	Gait training. Balance training	Stroke (N=2)
		Upper extremity therapeutic exer-	
		cise. Overall function task training	
[32]	Shoulder syndromes	Exercise rotator cuff protocol	Rotator cuff pathology (N=42)
[25]	Upper limb motor impairments	Exergames for patients with upper	Motor impairments (N=294)
	due to neurological conditions	and lower limb motor impairments	
[33]	Osteoarthritis	Exercises performed: hip exten-	Osteoarthritis (N=24)
		sion bent leg, knee flexion (leg	
		curl), strengthening hip extensors,	
		strengthening hip abductors, strain	
		front of the thigh, elongation of the	
		hip flexors	
[34]	Hand rehabilitation in stroke	Fugle Mayer exercise protocols:	Healthy subjects (N=4); subjects
		grasping, pinching, waving	with upper-limb weakness (N=3)
[35]	Physical activity	General ROM and muscular en-	Healthy subjects (N=10) (UI-
		durance exercises	PRMD dataset)
[36]	Pulmonary rehabilitation	Baduanjin exercises (N=8)	Chronic obstructive pulmonary
			disease (N=18); Healthy Subjects
			(N=5)
[37]	Low back pain rehabilitation	Back flexion/extension and rota-	Healthy subject(N=3)
		tions	
[38]	Hand rehabilitation in cerebral	Exercise performed to control	Young cerebral palsy (N=19)
	palsy	the game: wrist extension-open	
		fingers, wrist extension-closed	
		fingers, finger-thumb pinch, or	
		supination	
[39]	Balance	Balance static exercise: keep	Healthy subjects (N=21); Parkin-
		trunk upright or lean to	son's disease (N=20)
		left/right/front/back with different	
		angles, make a step forward and	
		steadily stepped off the board,	
		stand on one leg	

TABLE 4. Summary of information concerning the clinical application context. Abbreviations: ADL: activity of daily living, ROM: range of motion.

[40]	Stroke	Bilateral arm training rehabili- tation: bilateral shoulder flexion with both hands interlocked, wall push exercise, active scapular ex- ercise, towel slide exercise	Chronic stroke (N=23)
[41]	Cardiac rehabilitation	Local muscular endurance exer- cises: bicep curls, frontal raise, lat- eral raise, triceps extension right arm, trunk twist, squats, lunges- alternating sides, leg lateral raise, standing bicycle crunches	Healthy subjects (N=76)
[42]	Balance	Balance training exergame that elicits medio-lateral weight shifts from the participants	Healthy older adults (N=12)
[43]	Post-shoulder surgery	Shoulder and scapula rehabilita- tion movements	Healthy subjects (N=35)
[45]	Subacromial shoulder pain	Exercises for subacromial shoulder pain: shoulder abduction, shoulder flexion, wall slide, wall press and shoulder rotation	Patients with subacromial shoul- der pain (N=20)
[44]	Activity promotion in elderly	Sitting exercises (upper body twist, hip marching, arm raises, neck stretch), flexibility exercises (neck stretch, sideways bend) strength exercises (mini squats, sideways leg lift, leg extension.	Healthy subjects (N=11)

TABLE 4.	(Continued.) Summary of information concerning the clinical application context. Abbreviations: ADL: activity of daily living, ROM: range of
motion.	

		sideways leg lift, leg extension, bicep curls), balance exercises (simple grapevine, heel to toe walk)	
[46]	Activity promotion	General physical activities of daily life	Non-specific volunteers (N=375); healthy subjects (N=19)
[47]	Balance	Balance/agility-based tasks divided in sub-actions: squat, forward lunge, backward lunge	Parkinson's disease (N=35)
[48]	Patients with hip prosthesis	Hip rehabilitation exercise: one step forward, one step sideways and one step backward, with vari- ations	Healthy subjects (N=4)
[49]	Post-hospitalisation rehabilitation	Upper limb movements <i>via</i> exergame platform	Participants with different neuro- logical and orthopedic conditions (N=41)
[50]	Upper limb rehabilitation in stroke	Upper limb goal-directed ADL tasks to train, strength, ROM, functional capacity	Stroke (N=20); healthy subjects (N=10)
[51]	Breast cancer related lymphedema	Evidence-based exercise program to promote lymph flow for upper limbs	Healthy subjects (N=14)
[52]	Shoulder rehabilitation cuff- rotator problems	Evidence-based rotator cuff phys- iotherapy protocol: pendulum, ab- duction, forward elevation, in- ternal rotation, external rotation, trapezius extension, upright row	Healthy subjects (N=20)

[53]	Knee rehabilitation	Walking, knee flexion/extension exercise	Healthy subjects (N=6)
[54]	Knee rehabilitation	Knee exercises: the heel slide, the seated knee extension, the inner range quadriceps, the straight leg raise	Clinical subjects (N=44); Healthy subjects (N=10)
[55]	Hand rehabilitation	Seven hand exercise: extension- flexion, close and spread fingers, fingertip tap, radial-ulnar devia- tion, fingertip touch, palm rotation and finger mass extension-flexion	Healthy subject(N=17)
[56]	Physical activity	Squat	Healthy subject(N=4)
[57]	Shoulder rehabilitation	Shoulder flexion	Healthy subject(N=1)
[58]	Upper limb rehabilitation in stroke	Functional upper limb movements	Stroke (N=23); Healthy subjects (N=4)
[59]	Knee rehabilitation	Standing hamstring curl, straight leg raise exercises	Knee osteoarthritis (N=8)

TABLE 4.	(Continued.) Summary of information concerning the clinical application contex	kt. Abbreviations: ADL: activity of daily living, ROM: range of
motion.		

video and audio feedback during a serious game session [25], [26], [27], [38], [42], [49]; digital coach solutions that, *via* speech [29], [33], [47], sometimes involving social humanoid robot to encourage activity participation [44], or app-based text notifications [46], offer automated conversational interaction to replace some human care tasks (reminders and motivational messages for medication, nutrition, and exercise, routine condition checks and health maintenance based on personal monitoring data).

Over the past decade, significant efforts focused on developing unobtrusive, effective, and objective motion-modeling systems have been made, taking advantage of the progress made in sensor technology which has become more compact and more efficient [62]. In almost all the included works, motion capture technologies based on wearable sensors and optical sensor-based systems have been mainly individualized as useful systems to support the remote monitoring of physical rehabilitation tasks. Such technologies allow for providing crucial information about the patient's movement during physical exercise sessions by analyzing significant kinetic (force, torque, moments, *etc.*) and kinematic (angle, orientation, velocity, *etc.*) features collected in an unobtrusively way [63].

Wearable systems are based on compact, lightweight sensors directly mounted on the interesting body part to analyze, sometimes resulting in embedded smart textile systems such as garments and gloves [28], [34], [37] or wrist-worn devices such as smart bands and smartwatches [32], [40], [49], [52], [53], [57], or as the hardware part of immersive virtual reality systems (haptic interfaces and headset) [27]. The Inertial Measurement Unit (IMU) was the most common motion capture technology thanks to its low cost and wearability. IMUs combine linear acceleration from the accelerometer and the angular turning rates from gyroscopes resulting in accurate motion data. Moreover, IMUs coupled with different sensors such as optical fiber sensors [28] and flex and force resistive sensors [34] are useful to acquire more information about the amount of deflection or bending and the force applied during a rehabilitation task. Acceleration signals were also used and coupled to electrocardiogram (ECG) data in order to better recognize the patient's physical activity form [56].

Recently, the advancements in optical sensor-based motion capture technologies attracted researchers to perform unobtrusive motion analysis. The vision-based sensor technology uses the contactless approach for motion capture; at the same time, it provides reliable movement tracking without influencing its naturalness, extraction of kinetic and kinematic parameters, and accuracy in non-controlled environments [64]. The Microsoft Kinect camera was the vision-based system majorly used in motion tracking. Meanwhile, the Leap Motion system was more specifically applied in the field of hand rehabilitation [55].

Besides all these types of motion capture technologies, the recent developments in the computer vision field extended the use of low-cost RGB and depth sensors to perform motion analysis by means of human pose estimation algorithms based on deep learning frameworks [30], [33], [42]. Human pose estimation is a field of computer vision that aims to predict the poses of human bodies by extracting joints from images and videos for motion analysis [65]. Contrarily to wearable sensors, AI-based human motion modeling enables commercial systems equipped with a camera and low-cost hardware, such as tablets and smartphones, to perform inexpensive and unobtrusive home-based monitoring in patients' daily life [33], [35], [36].

C. MACHINE LEARNING METHODS

Machine learning is an artificial intelligence application that allows systems to automatically learn and improve from experience without being explicitly programmed to do so

Reference	Type of study (setting)	Implications	Drawbacks
[26]	Preliminary experimental study	Facilitate in-home balance	A limited sample of participants;
	(lab)	training by incorporating accurate	limited collected data focused
		feedback on weight-shifting	exclusively on sideways leaning
		performance	movements
[27]	Technical study (lab)	The method used only controllers	Limited sample of participants;
		and headset data of an intuitive	limited results generalizability due
		gaming system, making it ideal for	to data collected on subjects with
		at-home rehabilitation and evalua-	same clinical characteristics
		tion sessions	
[28]	Feasibility study (lab)	Optimized option for remote	Testing the whole system in real
		healthcare applications to identify	healthcare scenario
		activities and extract different	
		compact components integrated	
		into a usual clothing	
[29]	Feasibility study (lab)	The work contributes to the de-	More features should be used to
	i custoning study (nuo)	velopment of virtual coaches that	improve the characterization of
		help to achieve healthy aging by	complex exercise
		supporting regular daily exercise,	1
		improving adherence to the physi-	
		cal routine, and monitoring it	
[30]	Feasibility study (lab)	Monitor joint angular velocities	Angular velocity error between 15
		using video information without	to 20 rad/s
		wearable sensors	
[31]	Pilot study (lab)	Monitoring patients' performance	Limited sample; manual annotated
		during in-home rehabilitation ex-	dataset
		ercises	
[32]	Clinical longitudinal cohort study	Track shoulder physiotherapy par-	The accuracy of the system de-
	(lab, nome)	leated in an unobtrucive and easy	technology by petients
		to use way	technology by patients
[25]	Case study (lab)	Predict individualized profile set-	n r
		tings for patients with a certain	
		condition	
[33]	Clinical validation study (lab)	Digital therapeutic solution acces-	The sample was heterogeneous in
		sible to a broad patient population	terms of gender distribution and
		that provides feedback during ex-	localization of impairments; lim-
		ercise performance	ited numbers of raters for inter-
			rater agreements
[34]	Preliminary experimental study	An in-home inexpensive auto-	The differing hand sizes can affect
	(lab)	mated assessment system for re-	the input data
		habilitation of upper-extremity pa-	
[25]	Durlinging and include 1	tients with high recovery	
[35]	(lab)	henchmark dataset	n.r.
[26]	(Iau) Preliminary experimental study	System evaluation score correlated	Limited data
	(lab)	to clinician based scores	
[37]	Feasibility study (lab)	Method based on a small number	Limited sample size
		of stretch sensors to estimate the	
		spinal motion	

TABLE 5. Summary of studies characteristics, implications, and drawbacks. Abbreviations: Lab: laboratory, n.r.: not reported, ADL: activity of daily living.

[38]	Clinical study (home)	The classification methods facili- tated practice of targeted therapeu- tic gestures	Ground truths were established manually <i>via</i> synchronized video evaluation
[39]	Laboratory validation study (lab)	Accurate and quantitative bal- ance assessments based on val- idated clinical instrument mini- BEST test	Kinematic data were collected only on the frontal plane
[40]	Clinical prospective comparative study (home)	Facilitate participation in-home training and improve the func- tional outcomes	Limited sample size
[41]	Comparative study (lab)	High accuracy in exercise recogni- tion tasks	The tasks of exercise recognition and repetition counting was as- sessed in an offline mode with a windowing method
[42]	Comparative study (lab)	Accurate camera-based motion capture for balance training monitoring	Limited sample size; kinematic data were acquired only on the frontal plane
[43]	Feasibility study (lab)	Accurate segmentation of exercise repetitions to provide information on exercise performance	Data was collected on non-specific target population
[45]	Preliminary clinical validation study (lab)	Promote patients' engagement and personalized treatment strategies	System was evaluated in a super- vised and controlled scenario
[44]	Preliminary experimental study (lab)	Accurate techniques for remote exercise recognition	Data collected in supervised labo- ratory conditions
[46]	Design and validation study (lab, home)	A text-messaging-based mobile application to easily promote be- havioral change interventions	Along-term follow-up is required to clinically validate the tool
[47]	Feasibility study (lab)	Automatic monitoring and assis- tance tool for balance rehabilita- tion	The detection accuracy could de- grade when tracking more compli- cated movements
[48]	Preliminary experimental study (lab)	The system is composed of a user interface for the patients perform- ing the therapeutic physical exer- cises and another interface to en- able the health professionals to re- motely monitor the recovery pro- cess and update the rehabilitation program	Limited sample
[49]	Preliminary clinical validation study (lab)	Automatic evaluation of patient activity performance	Heterogeneous sample
[50]	Feasibility and usability study (lab)	Develop individualized therapy plans to maximize a patient's abil- ity to perform ADL and live inde- pendently	Sample characteristics limited to chronic stroke survivors with mild-to-moderate upper-limb motor impairments
[51]	Technical study (lab)	Correct errors in Kinect measure- ments due to self-occlusion	n.r.
[52]	Feasibility study (lab)	Objective measurement of therapy adherence in-home rehabilitation protocols	Data was collected on non-specific target population
[53]	Pilot study (lab)	Unobtrusive wearable system for monitoring daily life activities	Small sample size, laboratory evaluation setting
[54]	Case study (clin)	Application for real-time biofeed- back regarding exercise perfor- mance	Heterogeneous clinical character- istics and conditions

TABLE 5. (Continued.) Summary of studies characteristics, implications, and drawbacks. Abbreviations: Lab: laboratory, n.r.: not reported, ADL: activity of daily living.

[55]	Preliminary experimental stud	y Unobtrusive motion system for	n.r.
	(lab)	monitoring hand rehabilitation ex-	
		ercises	
[56]	Feasibility study (lab)	The fusion of the best classifica-	n.r.
		tion rate with high computational	
		efficacy makes the RF a reason-	
		able solution for embedded imple-	
		mentation on wearable devices	
[57]	Preliminary experimental stud	y Using device based only on ac-	Data collected in supervised labo-
	(lab)	celerometer data to monitoring	ratory conditions
		ROM	
[58]	Preliminary experimental stud	y Automatic assessment based on a	Small training set, heterogeneous
	(lab)	clinical validated method	clinical characteristics
[59]	Preliminary experimental stuc	y The system assesses the quality	Data collected in supervised labo-
	(lab)	of motion performed and provides	ratory conditions
		detailed feedback to the user	

TABLE 5. (Continued.) Summary of studies characteristics, implications, and drawbacks. Abbreviations: Lab: laboratory, n.r.: not reported, ADL: activity of daily living.



FIGURE 5. Categorization of machine learning algorithms. Abbreviations: LSTM: long short-term memory, MLP: multilayer perceptron, CNN: convolution neural network.

using previously selected features. Depending on scopes and tasks, machine learning algorithms can be divided into unsupervised and supervised learning. Unsupervised learning is well known for feature extraction, while supervised learning is suitable for predictive modeling by building relationships between features and the target of interest [66]. Retrieved machine learning algorithms are schematically presented in Figure 5.

In only one study that emerged, an unsupervised machine learning technique known as Self Organized Map (SOM) was used to individualize some clusters according to patients' demographic and clinical characteristics [25]. SOMs are artificial neural networks that generate a feature map to produce a low-dimensional and discretized representation of the input training samples for clustering, visualization, and classification scopes [67].

Almost all the articles involved supervised machinelearning techniques for classification tasks. Regression analysis modeling relationships between dependent and independent variables involved linear regression algorithms in analyzing and learning from the existing training data. Linear regression algorithms have been applied in fewer works aiming, for example, to reduce the system complexity of an exergame platform by estimating ground reaction forces from kinematic features [26], or to solve self-occlusion errors in motion analysis mapping the noisy Kinect-based measurements to a more accurate benchmark system measurements [51].

Random Forest (RF) and, more specifically, Decision Trees (DT) algorithms were the most used classifiers. DT algorithm was also used for a regression problem concerning the feasibility of using motion tracking data to predict joint angles and torques during an immersive virtual exergame [27]. DT is one of the oldest machine learning algorithms; it bases its decision logic on a tree-like architecture [68]. Its ease of interpreting and its rapid learning speed made it popular to use in the healthcare domain, especially in multiclass activity recognition problems in rehabilitation. The reason is that when examining the tree for a classification sample, the results of each node will provide relevant information to be inferred about its class. RFs are sets of randomized DTs combined using bagging techniques to reduce overfitting problems when the dataset is relatively large [69]. An RF model is particularly useful for categorizing classes of patients according to their clinical assessment characteristics [39]. Support Vector Machine (SVM) and K-Nearest Neighbor (KNN) were the other traditional supervised machine learning algorithms most used in selected studies, followed by logistic regression classifiers, Naive Bayes (NB) algorithm, and the Hidden Markov Model (HMM) algorithm. HMM is a Markov chain process that finds application in posture recognition and characterization of skeletal tracking spatial-temporal data as a parametric stochastic model [70]. HMM was applied as a probabilistic approach to model a given action into hidden states representing an arbitrary decomposition of the whole movement into succes-

sive phases. For example, an HMM algorithm was applied to classify the performed exercise into six possible correctness classes according to its execution [48]. Also, logistic regression was used for movement evaluation [34], [49] and activity monitoring [50]. Logistic regression is considered the baseline supervised machine learning algorithm to classify an observation, and its outputs can be easily interpreted as probabilities of the occurrence of a class [71]. In the case where authors handled multiple classes [56], a generalization of logistic regression was used: the softmax regression algorithm [72]. A more simplest approach used to make class-probability inference was the NB algorithm. The NB classifier is fast and easy to implement, resulting in particularly used for the implementation of a recommendations system. However, one of its biggest disadvantage is that the algorithm is based on Bayes' theorem that assumes independence between the features which are dependent in most of the real-life cases [73].

The KNN is another algorithm often applied in classification problems. KNN is a non-parametric classification algorithm known for its simplicity and effectiveness, used to classify data based on closest or neighboring training examples in each region [74]. The KNN classifier is based on distance metrics and was widely used in real-time applications for activity monitoring [29], [41], [45], [50], [52] as it is free from the underlying assumptions about the distribution of the dataset. Finally, Support Vector Machines (SVM) was the third most used traditional machine learning classifier applied for classification tasks in activity recognition [29], [31], [38], [41], [45], [52], movement classification [47]but also for clinical assessment scopes [46]. SVM algorithms present good generalization ability for sequential data structures and datasets that are not too large and for linear and non-linear problems.

Artificial neural networks were also a common choice among researchers interested in implementing solutions for monitoring and assisting decentralized physical rehabilitation. Artificial neural networks are a subset of machine learning algorithms inspired by neuroscientific studies concerning the functioning of the neurons' brain, whose branch is known as deep learning [75]. The recent advancements in deep learning, such as hybrid and lightweight deep neural networks, have been exploited to develop a framework for exercise performance evaluation and smart assistance for home-based rehabilitation sessions. Convolutional neural network (CNN) was the commonest architecture applied in selected papers. CNNs are designed to learn spatial hierarchies of features automatically and adaptively through backpropagation by using multiple building blocks, such as convolution layers, pooling layers, and fully connected layers [76]. The CNNs have been trained on large-scale datasets and on graph-structured data for motion analysis issue [35], [77]. CNNs achieved outstanding accuracies in the computer vision field for human detection and pose estimation that is useful to compute position and orientation data of interesting joints [33], [39], [42], [44], but also

for activity monitoring [32], [40], [41], [52] and movement evaluation [43]. Particularly CNN architectures such as ResNet [78] and AlexNet [79] achieved the highest accuracy respectively in pose estimation method for tracking human motion [42] and activity monitoring by processing some kinematic data [41]. Also, recurrent neural network (RNN) models achieved high accuracy in activity monitoring, which, thanks to the redundant connections between the network's neurons, can capture temporal dependencies in input data [26], [80]. More in detail, long short-term memory (LSTM), which is an RNN variant model [81], is particularly useful for analyzing temporal sequences between motion data frames [44]. The latest common deep learning approach among articles was the multilayer perceptron (MLP). MLP showed to achieve high accuracy above all in detecting specific movements [31]. MLP is a feed-forward fully connected network consisting of an input layer that receives data, an output layer that makes the decision or prediction about the input signal, and one or more hidden layers between these two that are considered the network's computational engine [82]. MLP, not requiring preliminary feature engineering, is considered easier to develop and apply in smart assistant applications than other deep learning methods. Also, a more simple learning algorithm, called extreme learning machine (ELM) that is composed of a single-hidden layer feed-forward neural networks [83], was able to achieve high accuracy in multi-class classification task in work [58].

D. CLINICAL APPLICATIONS

Besides technical requirements, it is fundamental to integrate clinical knowledge into technologies to be used at home in a minimally supervised way. Hence, the role of AI in clinical practice should be to provide a combination of medical, psychological, and technical knowledge in the form of embedded algorithms analyzing and processing online the data generated by digital technologies. AI is expected to play a crucial role in clinical decision-making, the online adaptation of therapy exercises, and the monitoring of progress through the extraction of validated assessment scores [84]. All the information regarding the clinical application context is reported in Table 3. Based on the reviewed papers, three main approaches have been introduced to manage the remote monitoring and evaluation of physical rehabilitation therapies by means of machine learning applications. These approaches could be categorized as activity recognition, movement classification, and clinical functional assessment.

Machine learning algorithms address activity recognition tasks aimed to identify specific rehabilitation actions to remotely track patients' adherence to the prescribed therapy and sometimes to measure the treatment outcomes based on patients' activity in real-world life [85]. Most of the reviewed papers covered the problem of recognizing activities for specific body parts like strength training exercises for upper and lower limb [29], [41], [44], [56], exercises to improve range of motion [38], flexibility and balance exercises [37], [44], specific rehabilitation exercise for upper limb and lower impaired limb in stroke [31], [40], [55], and shoulder impairments [32], [45], [52]. Other articles monitored activities of daily living (ADL), detecting general activities such as standing, sitting, squatting, walking, and running useful for health promotion programs in the elderly [28], [44], and other routine activities involving upper limb goal-directed tasks in stroke rehabilitation [50].

Movement classification approaches aimed to assess the quality of rehabilitation exercises performed remotely. Algorithms addressing movement classification tasks have been used to evaluate exercise performance in terms of the well and poorly-executed tasks. Some approaches relied on classifying the movement into discrete score classes based on clinician rating, such as low, moderate, and high performance of hand movements in stroke rehabilitation [34], [58] and well or badly performed upper limb movements while playing exergame [49]. Always based on clinical expertise knowledge, other approaches classify movements in correct or erroneous execution classes defined by a set of pre-established rules [36], [47], [48], [54]. For example, a set of predefined rules was used to label correct and compensatory movements in different possible classes to provide real-time feedback during remote hip rehabilitation exercises [48]. Noteworthy, another approach based on the movement classification task was applied to develop an accurate segmentation of exercise repetitions to evaluate the patient's performance in a detailed way [43], [59].

The last interesting approach in reviewed papers aimed to manage rehabilitation interventions through preventive and personalized solutions by predicting the patient's clinical functional status. A personalized exergame based on classifying the user profile according to related impairment characteristics was introduced to adapt therapeutic settings during a motor neurorehabilitation session [25]. An accurate and quantitative balance assessment based on a validated clinical instrument mini-BEST test was proposed to recognize the level of patient impairment and predict the fall risk [39]. Finally, a multi-class classifier was used to separate new users into three groups based on their initial activity level (vigorous, mild, and sedentary users) to develop a conversational agent-assisted health coaching system to support behavioral change *via* personalized recommendations [46].

E. LIMITATIONS

The authors of reviewed articles evidenced and reported some drawbacks (see Table 4). They refer to issues regarding the size and characteristics of the samples and the low generalizability of results in a relevant operational environment.

In general, the data were collected from a limited sample composed mainly of healthy subjects that did not reflect the pathological behavior of the end-user population target for which the technologies have been developed. Few systems have been tested and validated on relevant clinical population targets: stroke [31], [40], [50], [58], Parkinson's disease [39], [47], cerebral palsy [38], patients with shoulder

VOLUME 11, 2023

impairments [32], [45], osteoarthritis [33], [59], chronic obstructive pulmonary disease [36]. Other studies recruited subjects with various motor impairments related to orthopedic and neurological conditions [25], [33], [49], [54], thus addressing an enormous heterogeneity in collected data.

Other problems concerning the generalizability of results have been addressed as related to the reproducibility of the data collected in limited or different conditions (such as settings, users, and operators) [26], [27], [28], [32], [34], [36], [39], [42], [46], [53], [57], [59].

Considering the technology readiness level (TRL) of all the reviewed solutions, the TRL generally spans from level 3 (experimental proof of concept) to level 4 (technology validated in the laboratory) [86]. A very minority of studies validated and demonstrated the application of proposed AI-based solutions in relevant operation environments such as remote conditions and home-based settings [32], [38], [40], [46]. Finally, analyzing reviewed articles for SJR quartile quality metrics emerged that high-quality research was conducted for activity recognition and clinical status prediction applications as expected. However, future efforts are needed to increase the quality of the research interested in solving rehabilitation issues by the means of movement classification applications (see Figure 4).

IV. DISCUSSION

This work presented a systematic review of machine learning methods and applications in the physical rehabilitation context to support the future challenge of making remote services accessible. During the last decade (from 2010 on), several system technologies coupled with machine learning methods have been applied to monitor and assess rehabilitation issues. A total of 519 publications were retrieved by a systematic search strategy performed in PubMed, IEEE Xplore, and Scopus databases. After the full-text screening, 35 articles that respected eligibility criteria were included in the data extraction process.

The study aimed to report the status of innovative developments obtained in the field of AI-based machine learning solutions supporting decentralized rehabilitation services. Study characteristics related to the type of system used, machine learning techniques and methods applied, information about the clinical application field evidencing rehabilitation scopes, the population target, and participants enrolled for validation issues were highlighted. Moreover, limitations and challenges encountered by researchers and future directions concerning the role of AI-based solutions in remote rehabilitation services were reported.

The real challenge of artificial intelligence applications resulted in the incorporation of clinical skills and knowledge in order to assist and manage patients in a minimally supervised and more efficient way.

Emerging AI-based computer vision approaches, which currently represent an intensive research topic, are promising for rehabilitation applications to provide home-based, inexpensive, and unobtrusive monitoring of patients [87]. Unobtrusive and wearable technologies coupled with algorithms that recognize performed activities will make the remote evaluation of patient's adherence to prescribed exercise affordable, which until now was based on self-reported measures by subjects [17]. Assessment-driven therapy adaptation algorithms integrated into exergame solutions should ensure that the physical therapy remains at an optimal level of challenge to maintain motivation for long-term therapy programs necessary in chronic conditions [88]. Moreover, movement classification algorithms integrated into digital coach systems and socially assistive robots could support real-time evaluations and personalized feedback. They provide emotional support to increase engagement during home-based rehabilitation therapy in clinical populations characterized by fragile psychophysical conditions [89], [90].

Gradual steps are necessary to introduce AI applications to support rehabilitation services at a distance. Firstly, AI-based monitoring systems integrated into a traditional videoconferencing application could be the way forward to a data-driven approach that helps therapists plan personalized and participatory programs during remote rehabilitation visits and sessions. Then, AI-based applications for automatic monitoring and assistance could be adopted to solve issues in long-term rehabilitation programs (to the maintenance of therapy outcomes and the overload of clinic facilities for routine practices). Furthermore, AI-based applications could also allow engaging a wide range of individuals in their home environments to easier extend preventive rehabilitation programs in pre-morbidities conditions.

Besides the potentiality of AI-based solutions to support the remote monitoring and evaluation of physical therapy programs, further developments of underlying algorithms and validation of methods will be required for broader adoption. Future studies must focus on testing AI-based applications in relevant operational environments, such as in remote conditions and home-based settings. Clinical trials comparing the effects on specific population targets of such emerging solutions versus traditional rehabilitation approaches are required to lead AI-based applications to clinical acceptance. Social and ethical considerations will also need to be considered as AI-based applications will change the paradigm of interactions occurring between healthcare professionals and patients in decentralized contexts.

V. CONCLUSION

This work outlined the potential applicability of machine learning methods within decentralized rehabilitation services.

The systematic review approach revealed the state-ofthe-art AI-based machine learning solutions implemented to assist and manage remote rehabilitation procedures during the last decade. Characteristics concerning systems and machine learning deployed in the rehabilitation context and their implications to support the delivery of therapies from a distance have been evidenced and discussed.

The above discussion suggests that AI-based solutions offer contributions in the form of quality improvement and

enhancement of existing practice, as well as support new models of care based on decentralized services.

The review and recommendations provided in this paper aim to guide the design of the next generation of AI-aided rehabilitation services, their validation, and their translation to clinical practice. In the future, AI is expected to increasingly play a crucial role in driving the clinic-centered model toward a decentralized healthcare model that will overcome challenges of health service delivery due to time, distance, and logistic issues, enabling cost-effectiveness and better access to long-term therapies.

VI. CONFLICTS OF INTERESTS

The authors have no conflicts of interest to declare. All co-authors have seen and agree with the contents of the manuscript and there is no financial interest to report. We certify that the submission is original work and is not under review at any other publication.

REFERENCES

- The 2015 Ageing Report : Economic and Budgetary Projections for the 28 EU Member States (2013–2060), Publications Office, European Commission, Directorate-General for Economic, Financial Affairs, and Economic Policy Committee, Strasbourg, France, 2015.
- [2] R. E. Jordan, P. Adab, and K. K. Cheng, "COVID-19: Risk factors for severe disease and death," *Brit. Med. J.*, vol. 368, Mar. 2020, Art. no. m1198.
- [3] V. Caso and A. Federico, "No lockdown for neurological diseases during COVID19 pandemic infection," *Neurol. Sci., Off. J. Italian Neurol. Soc. Italian Soc. Clin. Neurophysiol.*, vol. 41, no. 5, pp. 999–1001, 2020.
- [4] A. Turolla, G. Rossettini, A. Viceconti, A. Palese, and T. Geri, "Musculoskeletal physical therapy during the COVID-19 pandemic: Is telerehabilitation the answer?" *Phys. Therapy*, vol. 100, no. 8, pp. 1260–1264, Aug. 2020.
- [5] S. Nicholson, F. F. Sniehotta, F. van Wijck, C. A. Greig, M. Johnston, M. E. T. Mcmurdo, M. Dennis, and G. E. Mead, "A systematic review of perceived barriers and motivators to physical activity after stroke," *Int. J. Stroke*, vol. 8, no. 5, pp. 357–364, Jul. 2013.
- [6] J. J. Daly, J. P. McCabe, J. Holcomb, M. Monkiewicz, J. Gansen, and S. Pundik, "Long-dose intensive therapy is necessary for strong, clinically significant, upper limb functional gains and retained gains in severe/moderate chronic stroke," *Neurorehabil. Neural Repair*, vol. 33, no. 7, pp. 523–537, Jul. 2019.
- [7] N. S. Ward, F. Brander, and K. Kelly, "Intensive upper limb neurorehabilitation in chronic stroke: Outcomes from the queen square programme," *J. Neurol., Neurosurg. Psychiatry*, vol. 90, no. 5, pp. 498–506, May 2019.
- [8] E. N. Mayo, S. Wood-Dauphinee, R. Côté, D. Gayton, J. Carlton, J. Buttery, and R. Tamblyn, "There's no place like home," *Stroke*, vol. 31, no. 5, pp. 1016–1023, 2000.
- [9] S. Hillier and G. Inglis-Jassiem, "Rehabilitation for community-dwelling people with stroke: Home or centre based? A systematic review," *Int. J. Stroke*, vol. 5, no. 3, pp. 178–186, Jun. 2010.
- [10] N. E. Mayo, "Stroke rehabilitation at home," *Stroke*, vol. 47, no. 6, pp. 1685–1691, Jun. 2016.
- [11] A Health Telematics Policy in Support of Who's Health-for-All Strategy for Global Health Development: Report of the Who Group Consultation on Health Telematics, Switzerland WHO Group Consultation on Health Telematics: Geneva, Switzerland, Dec. 1998.
- [12] R. Argent, A. Daly, and B. Caulfield, "Patient involvement with homebased exercise programs: Can connected health interventions influence adherence?" *JMIR mHealth uHealth*, vol. 6, no. 3, p. e47, Mar. 2018.
- [13] L. T. Car, D. A. Dhinagaran, B. M. Kyaw, T. Kowatsch, S. Joty, Y.-L. Theng, and R. Atun, "Conversational agents in health care: Scoping review and conceptual analysis," *J. Med. Internet Res.*, vol. 22, no. 8, Aug. 2020, Art. no. e17158.

- [14] C. M. Kanzler, M. D. Rinderknecht, A. Schwarz, I. Lamers, C. Gagnon, J. P. O. Held, P. Feys, A. R. Luft, R. Gassert, and O. Lambercy, "A datadriven framework for selecting and validating digital health metrics: Usecase in neurological sensorimotor impairments," *NPJ Digit. Med.*, vol. 3, no. 1, p. 80, May 2020.
- [15] V. Nangalia, D. R. Prytherch, and G. B. Smith, "Health technology assessment review: Remote monitoring of vital signs-current status and future challenges," *Crit. Care*, vol. 14, no. 5, p. 233, 2010.
- [16] T. Davergne, A. Rakotozafiarison, H. Servy, and L. Gossec, "Wearable activity trackers in the management of rheumatic diseases: Where are we in 2020?" *Sensors*, vol. 20, no. 17, p. 4797, Aug. 2020.
- [17] I. Boukhennoufa, X. Zhai, V. Utti, J. Jackson, and K. D. McDonald-Maier, "Wearable sensors and machine learning in post-stroke rehabilitation assessment: A systematic review," *Biomed. Signal Process. Control*, vol. 71, Jan. 2022, Art. no. 103197.
- [18] N. K. Mangal and A. K. Tiwari, "A review of the evolution of scientific literature on technology-assisted approaches using RGB-D sensors for musculoskeletal health monitoring," *Comput. Biol. Med.*, vol. 132, May 2021, Art. no. 104316.
- [19] Y. Liao, A. Vakanski, M. Xian, D. Paul, and R. Baker, "A review of computational approaches for evaluation of rehabilitation exercises," *Comput. Biol. Med.*, vol. 119, Apr. 2020, Art. no. 103687.
- [20] M. Jones, G. Collier, D. J. Reinkensmeyer, F. DeRuyter, J. Dzivak, D. Zondervan, and J. Morris, "Big data analytics and sensor-enhanced activity management to improve effectiveness and efficiency of outpatient medical rehabilitation," *Int. J. Environ. Res. Public Health*, vol. 17, no. 3, p. 748, Jan. 2020.
- [21] H. Snyder, "Literature review as a research methodology: An overview and guidelines," J. Bus. Res., vol. 104, pp. 333–339, Nov. 2019.
- [22] M. Vaismoradi, H. Turunen, and T. Bondas, "Content analysis and thematic analysis: Implications for conducting a qualitative descriptive study," *Nursing Health Sci.*, vol. 15, no. 3, pp. 398–405, Sep. 2013.
- [23] J. Higgins, J. Thomas, J. Chandler, M. Cumpston, T. Li, M. Page, and V. Welch, *Cochrane Handbook for Systematic Reviews of Interventions*. Chichester, U.K.: Wiley, Sep. 2019.
- [24] D. Moher, A. Liberati, J. Tetzlaff, and D. G. Altman, "Preferred reporting items for systematic reviews and meta-analyses: The PRISMA statement," *PLoS Med.*, vol. 6, pp. 1–6, Jul. 2009.
- [25] N. Gharaei, W. Ismail, C. Grosan, and R. Hendradi, "Optimizing the setting of medical interactive rehabilitation assistant platform to improve the performance of the patients: A case study," *Artif. Intell. Med.*, vol. 120, Oct. 2021, Art. no. 102151.
- [26] E. K. Vonstad, K. Bach, B. Vereijken, X. Su, and J. H. Nilsen, "Performance of machine learning models in estimation of ground reaction forces during balance exergaming," *J. NeuroEng. Rehabil.*, vol. 19, no. 1, pp. 1–12, Dec. 2022.
- [27] M. O. Powell, A. Elor, A. Robbins, S. Kurniawan, and M. Teodorescu, "Predictive shoulder kinematics of rehabilitation exercises through immersive virtual reality," *IEEE Access*, vol. 10, pp. 25621–25632, 2022.
- [28] L. Avellar, C. S. Filho, G. Delgado, A. Frizera, E. Rocon, and A. Leal-Junior, "AI-enabled photonic smart garment for movement analysis," *Sci. Rep.*, vol. 12, no. 1, p. 4067, Mar. 2022.
- [29] S. García-de-Villa, D. Casillas-Pérez, A. Jiménez-Martín, and J. J. García-Domínguez, "Simultaneous exercise recognition and evaluation in prescribed routines: Approach to virtual coaches," *Expert Syst. Appl.*, vol. 199, Aug. 2022, Art. no. 116990.
- [30] K. S. Kumar, A. Jamsrandorj, J. Kim, and K.-R. Mun, "Prediction of lower limb kinematics from vision-based system using deep learning approaches," in *Proc. 44th Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.* (*EMBC*), Jul. 2022, pp. 177–181.
- [31] V. C. E. Bernal, S. V. Hiremath, B. Wolf, B. Riley, R. J. Mendonca, and M. J. Johnson, "Classifying and tracking rehabilitation interventions through machine-learning algorithms in individuals with stroke," *J. Rehabil. Assistive Technol. Eng.*, vol. 8, Jan. 2021, Art. no. 205566832110446.
- [32] D. Burns, P. Boyer, H. Razmjou, R. Richards, and C. Whyne, "Adherence patterns and dose response of physiotherapy for rotator cuff pathology: Longitudinal cohort study," *JMIR Rehabil. Assistive Technol.*, vol. 8, no. 1, Mar. 2021, Art. no. e21374.
- [33] J. T. Biebl, M. Rykala, M. Strobel, P. K. Bollinger, B. Ulm, E. Kraft, S. Huber, and A. Lorenz, "App-based feedback for rehabilitation exercise correction in patients with knee or hip osteoarthritis: Prospective cohort study," *J. Med. Internet Res.*, vol. 23, no. 7, Jul. 2021, Art. no. e26658.
- VOLUME 11, 2023

- [34] H. Sarwat, H. Sarwat, S. A. Maged, T. H. Emara, A. M. Elbokl, and M. I. Awad, "Design of a data glove for assessment of hand performance using supervised machine learning," *Sensors*, vol. 21, no. 21, p. 6948, Oct. 2021.
- [35] C. Du, S. Graham, C. Depp, and T. Nguyen, "Assessing physical rehabilitation exercises using graph convolutional network with self-supervised regularization," in *Proc. 43rd Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.* (*EMBC*), Nov. 2021, pp. 281–285.
- [36] L. Guo, B. Zhang, Z. Yang, X. Wang, and J. Wang, "Motor function evaluation system based on machine vision for COPD patients," in *Proc. IEEE 3rd Int. Conf. Civil Aviation Saf. Inf. Technol. (ICCASIT)*, Oct. 2021, pp. 1136–1139.
- [37] J. Chen, J. Caviedes, and B. Li, "Classification of single-axis spinal motion using a wearable system of stretch sensors for at-home physical therapy," in *Proc. 43rd Annu. Int. Conf. IEEE Eng. Med. Biol. Soc. (EMBC)*, Nov. 2021, pp. 7404–7407.
- [38] A. Macintosh, N. Vignais, E. Desailly, E. Biddiss, and V. Vigneron, "A classification and calibration procedure for gesture specific homebased therapy exercise in young people with cerebral palsy," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 29, pp. 144–155, 2021.
- [39] W. Wei, C. Mcelroy, and S. Dey, "Using sensors and deep learning to enable on-demand balance evaluation for effective physical therapy," *IEEE Access*, vol. 8, pp. 99889–99899, 2020.
- [40] S. H. Chae, Y. Kim, K.-S. Lee, and H.-S. Park, "Development and clinical evaluation of a web-based upper limb home rehabilitation system using a smartwatch and machine learning model for chronic stroke survivors: Prospective comparative study," *JMIR mHealth uHealth*, vol. 8, no. 7, Jul. 2020, Art. no. e17216.
- [41] G. Prabhu, N. E. O'Connor, and K. Moran, "Recognition and repetition counting for local muscular endurance exercises in exercise-based rehabilitation: A comparative study using artificial intelligence models," *Sensors*, vol. 20, no. 17, p. 4791, Aug. 2020.
- [42] E. K. Vonstad, X. Su, B. Vereijken, K. Bach, and J. H. Nilsen, "Comparison of a deep learning-based pose estimation system to marker-based and Kinect systems in exergaming for balance training," *Sensors*, vol. 20, no. 23, p. 6940, Dec. 2020.
- [43] L. Brennan, A. Bevilacqua, T. Kechadi, and B. Caulfield, "Segmentation of shoulder rehabilitation exercises for single and multiple inertial sensor systems," *J. Rehabil. Assistive Technol. Eng.*, vol. 7, Jan. 2020, Art. no. 205566832091537.
- [44] E. Martinez-Martin and M. Cazorla, "A socially assistive robot for elderly exercise promotion," *IEEE Access*, vol. 7, pp. 75515–75529, 2019.
- [45] L. Bavan, K. Surmacz, D. Beard, S. Mellon, and J. Rees, "Adherence monitoring of rehabilitation exercise with inertial sensors: A clinical validation study," *Gait Posture*, vol. 70, pp. 211–217, May 2019.
- [46] A. Fadhil, Y. Wang, and H. Reiterer, "Assistive conversational agent for health coaching: A validation study," *Methods Inf. Med.*, vol. 58, no. 1, pp. 9–23, Jun. 2019.
- [47] W. Wei, C. McElroy, and S. Dey, "Towards on-demand virtual physical therapist: Machine learning-based patient action understanding, assessment and task recommendation," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 27, no. 9, pp. 1824–1835, Sep. 2019.
- [48] Y. Rybarczyk, C. Cointe, T. Gonçalves, V. Minhoto, J. K. Deters, S. Villarreal, A. A. Gonzalo, J. Baldeón, and D. Esparza, "On the use of natural user interfaces in physical rehabilitation: A web-based application for patients with hip prosthesis," *J. Sci. Technol. Arts*, vol. 10, no. 2, pp. 15–24, May 2018.
- [49] M. Morando, S. Ponte, E. Ferrara, and S. Dellepiane, "Definition of motion and biophysical indicators for home-based rehabilitation through serious games," *Information*, vol. 9, no. 5, p. 105, May 2018.
- [50] S. I. Lee, "Enabling stroke rehabilitation in home and community settings: A wearable sensor-based approach for upper-limb motor training," *IEEE J. Transl. Eng. Health Med.*, vol. 6, pp. 1–11, 2018.
- [51] A.-T. Chiang, Q. Chen, Y. Wang, and M. R. Fu, "Kinect-based in-home exercise system for lymphatic health and lymphedema intervention," *IEEE J. Transl. Eng. Health Med.*, vol. 6, pp. 1–13, 2018.
- [52] D. M. Burns, N. Leung, M. Hardisty, C. M. Whyne, P. Henry, and S. McLachlin, "Shoulder physiotherapy exercise recognition: Machine learning the inertial signals from a smartwatch," *Physiolog. Meas.*, vol. 39, no. 7, Jul. 2018, Art. no. 075007.

- [53] M. Gholami, A. Ejupi, A. Rezaei, A. Ferrone, and C. Menon, "Estimation of knee joint angle using a fabric-based strain sensor and machine learning: A preliminary investigation," in *Proc. 7th IEEE Int. Conf. Biomed. Robot. Biomechatronics (Biorob)*, Aug. 2018, pp. 589–594.
- [54] A. Bevilacqua, B. Huang, R. Argent, B. Caulfield, and T. Kechadi, "Automatic classification of knee rehabilitation exercises using a single inertial sensor: A case study," in *Proc. IEEE 15th Int. Conf. Wearable Implant. Body Sensor Netw. (BSN)*, Mar. 2018, pp. 21–24.
- [55] W.-J. Li, C.-Y. Hsieh, L.-F. Lin, and W.-C. Chu, "Hand gesture recognition for post-stroke rehabilitation using leap motion," in *Proc. Int. Conf. Appl. Syst. Innov. (ICASI)*, May 2017, pp. 386–388.
- [56] E. Kántoch, D. Grochala, M. Kajor, and D. Kucharski, "The prototype of wearable sensors system for supervision of patient rehabilitation using artificial intelligence methods," in *Innovations in Biomedical Engineering* (Advances in Intelligent Systems and Computing), vol. 623. Cham, Switzerland: Springer, 2018.
- [57] K. Ongvisatepaiboon, J. H. Chan, and V. Vanijja, "Smartphone-based tele-rehabilitation system for frozen shoulder using a machine learning approach," in *Proc. IEEE Symp. Ser. Comput. Intell.*, Dec. 2015, pp. 811–815.
- [58] L. Yu, J.-P. Wang, Q. Fang, and Y. Wang, "Brunnstrom stage automatic evaluation for stroke patients using extreme learning machine," in *Proc. IEEE Biomed. Circuits Syst. Conf. (BioCAS)*, Nov. 2012, pp. 380–383.
- [59] P. E. Taylor, G. J. M. Almeida, J. K. Hodgins, and T. Kanade, "Multi-label classification for the analysis of human motion quality," in *Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, Aug. 2012, pp. 2214–2218.
- [60] D. Theodoros and T. Russell, "Telerehabilitation: Current perspectives," Stud. Health Technol. Informat., vol. 131, pp. 191–209, Feb. 2008.
- [61] C. Kuziemsky, A. J. Maeder, O. John, S. B. Gogia, A. Basu, S. Meher, and M. Ito, "Role of artificial intelligence within the telehealth domain: Official 2019 yearbook contribution by the members of IMIA telehealth working group," *Yearbook Med. Informat.*, vol. 28, no. 1, pp. 35–40, Aug. 2019.
- [62] S. Majumder, T. Mondal, and M. Deen, "Wearable sensors for remote health monitoring," *Sensors*, vol. 17, no. 12, p. 130, Jan. 2017.
- [63] C. Mennella, S. Alloisio, A. Novellino, and F. Viti, "Characteristics and applications of technology-aided hand functional assessment: A systematic review," *Sensors*, vol. 22, no. 1, p. 199, Dec. 2021.
- [64] T. B. Moeslund and E. Granum, "A survey of computer vision-based human motion capture," *Comput. Vis. Image Understand.*, vol. 81, no. 3, pp. 231–268, Mar. 2001.
- [65] B. Debnath, M. O'Brien, M. Yamaguchi, and A. Behera, "A review of computer vision-based approaches for physical rehabilitation and assessment," *Multimedia Syst.*, vol. 28, no. 1, pp. 209–239, Feb. 2022.
- [66] F. Jiang, Y. Jiang, H. Zhi, Y. Dong, H. Li, S. Ma, Y. Wang, Q. Dong, H. Shen, and Y. Wang, "Artificial intelligence in healthcare: Past, present and future," *Brit. Med. J.*, vol. 2, no. 4, 2017, Art. no. e000101.
- [67] D. Miljkovic, "Brief review of self-organizing maps," in Proc. 40th Int. Conv. Inf. Commun. Technol., Electron. Microelectron. (MIPRO), May 2017, pp. 1061–1066.
- [68] S. Kotsiantis, "Decision trees: A recent overview," Artif. Intell. Rev., vol. 39, no. 4, pp. 261–283, 2013.
- [69] A. Parmar, R. Katariya, and V. Patel, A Review on Random Forest: An Ensemble Classifier. Cham, Switzerland: Springer, Jan. 2019, pp. 758–763.
- [70] J. F.-S. Lin and D. Kulic, "Segmenting human motion for automated rehabilitation exercise analysis," in *Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, Aug. 2012, pp. 4–2881.
- [71] Y. Andrew Ng and I. M. Jordan, "On discriminative vs. generative classifiers: A comparison of logistic regression and naive Bayes," in *Proc. Adv. Neural Inf. Process. Syst. (NIPS)*, Cambridge, MA, USA: MIT Press, 2001, pp. 841–848.
- [72] W. H. Greene, Econometric Analysis. London, U.K.: Pearson, 2003.
- [73] P. Cichosz, Naïve Bayes Classifier. Hoboken, NJ, USA: Wiley, 2015, pp. 118–133.
- [74] K. Taunk, S. De, S. Verma, and A. Swetapadma, "A brief review of nearest neighbor algorithm for learning and classification," in *Proc. Int. Conf. Intell. Comput. Control Syst. (ICCS)*, May 2019, pp. 1255–1260.
- [75] J. Heaton, "Ian Goodfellow, Yoshua Bengio, and Aaron Courville: Deep learning: The MIT Press, 2016, 800, pp. ISBN: 0262035618," *Genetic Program. Evolvable Mach.*, vol. 19, nos. 1–2, pp. 305–307, Jun. 2018.

- [76] R. Yamashita, M. Nishio, R. K. G. Do, and K. Togashi, "Convolutional neural networks: An overview and application in radiology," *Insights Imag.*, vol. 9, pp. 611–629, Jun. 2018.
- [77] T. N. Kipf and M. Welling, "Semi-supervised classification with graph convolutional networks," 2016, arXiv:1609.02907.
- [78] K. He, X. Zhang, S. Ren, and J. Sun, "Deep residual learning for image recognition," in *Proc. IEEE Conf. Comput. Vis. Pattern Recognit. (CVPR)*, Jun. 2016, pp. 770–778.
- [79] A. Krizhevsky, I. Sutskever, and G. E. Hinton, "ImageNet classification with deep convolutional neural networks," in *Adv. Neural Inf. Process. Syst.*, vol. 25, F. Pereira, C. J. Burges, L. Bottou, and K. Q. Weinberger, Eds. Red Hook, NY, USA: Curran Associates, 2012, pp. 200–209.
- [80] V. A, F. Jm, and L. S, "Mathematical modeling and evaluation of human motions in physical therapy using mixture density neural networks," *J. Physiotherapy Phys. Rehabil.*, vol. 1, no. 4, pp. 1–11, 2016.
- [81] G. Lefebvre, S. Berlemont, F. Mamalet, and C. Garcia, "BLSTM-RNN based 3D gesture classification," in *Proc. ICANN*, 2013, pp. 381–388.
- [82] J. Han, J. Pei, and M. Kamber, *Data Mining: Concepts and Techniques*. Amsterdam, The Netherlands: Elsevier, 2011.
- [83] G.-B. Huang, Q.-Y. Zhu, and C.-K. Siew, "Extreme learning machine: Theory and applications," *Neurocomputing*, vol. 70, nos. 1–3, pp. 489–501, 2006.
- [84] O. Lambercy, R. Lehner, K. Chua, S. K. Wee, D. K. Rajeswaran, C. W. K. Kuah, W. T. Ang, P. Liang, D. Campolo, A. Hussain, G. Aguirre-Ollinger, C. Guan, C. M. Kanzler, N. Wenderoth, and R. Gassert, "Neurorehabilitation from a distance: Can intelligent technology support decentralized access to quality therapy?" *Frontiers Robot. AI*, vol. 8, May 2021, Art. no. 612415.
- [85] International Classification of Functioning, Disability and Health : ICF. Geneva, Switzerland: World Health Organization, 2001.
- [86] S. Krahn, K. Gerdes, and H. Sutter, "Department of energy technology readiness assessments-process guide and training plan," Dept. Energy, Washington, DC, USA, Tech. Rep. DOE G 413.3-4 10-12-09, 2008.
- [87] T. Hellsten, J. Karlsson, M. Shamsuzzaman, and G. Pulkkis, "The potential of computer vision-based marker-less human motion analysis for rehabilitation," *Rehabil. Process Outcome*, vol. 10, Jan. 2021, Art. no. 117957272110223.
- [88] C. Giang, E. Pirondini, N. Kinany, C. Pierella, A. Panarese, M. Coscia, J. Miehlbradt, C. Magnin, P. Nicolo, A. Guggisberg, and S. Micera, "Motor improvement estimation and task adaptation for personalized robot-aided therapy: A feasibility study," *Biomed. Eng. OnLine*, vol. 19, no. 1, pp. 1–25, Dec. 2020.
- [89] P. Tropea, H. Schlieter, I. Sterpi, E. Judica, K. Gand, M. Caprino, I. Gabilondo, J. Gómez-Esteban, S. Busnatu, C. Sinescu, S. Kyriazakos, S. Anwar, and M. Corbo, "Scoping review on virtual coaching in medical care: Rehabilitation, the great absentee (preprint)," *J. Med. Internet Res.*, vol. 21, Nov. 2018, Art. no. e12805.
- [90] M. Lee, D. Siewiorek, A. Smailagic, A. Bernardino, and S. B. I. Badia, Enabling AI and Robotic Coaches for Physical Rehabilitation Therapy: Iterative Design and Evaluation With Therapists and Post-Stroke Survivors. Germany: Springer-Verlag, Jun. 2021.



CIRO MENNELLA was born in Torre del Greco, Napoli, Italy, in 1993. He received the B.S. degree in physiotherapy from the University of Naples Federico II, Napoli, in 2017, and the M.S. degree in rehabilitation sciences from Vita-Salute San Raffaele University, Milan, Italy, in 2019. He is currently pursuing the Ph.D. degree in artificial intelligence on health and life sciences with Campus Bio-Medico University, Rome, Italy, in collaboration with the Institute for High-Performance

Computing and Networking (ICAR), Research National Council (CNR), Genova, Italy. From 2020 to 2021, he was a Research Fellow with the Institute of Biophysics (IBF), CNR. His research interests include validation of new technologies for data-driven therapy approaches in rehabilitation medicine, AI for disease diagnosis and patient monitoring, machine learning for patient risk stratification, and augmenting clinical workflows.



UMBERTO MANISCALCO received the degree in electrical engineering and the Ph.D. degree in electrical, communication, and computer engineering from the University of Palermo, in November 1994 and December 1998, respectively. He has been an Applied Research Scientist (Technologist) with the High-Performance Computing and Network Institute, National Research Council, Italy, since December 2001. He is currently heading the Human–Robot Interaction Group. Since 2000,

he has been teaching undergraduate and graduate-level courses with the Architecture, Engineering, and Communication Faculty. He has participated in several local, national, and international research projects. He is the author or coauthor of over 70 scientific publications. He is/was a member of several scientific societies. He is/was a member of the AIxIA, SIREN, IMEKO, International Neural Network Society, European Neural Network Society, Japanese Neural Network Society, and Biologically Inspired Cognitive Architectures Society.



MASSIMO ESPOSITO received the M.Sc. degree (cum laude) in computer science engineering from the University of Naples Federico II, in March 2004, the master's degree (Hons.), named European Master on Critical Networked Systems, in December 2007, and the Ph.D. degree in information technology engineering from the University of Naples, Parthenope, in April 2011. Since 2012, he has been a Contract Professor of informatics with the Faculty of Engineering, University

of Naples Federico II. Since 2016, he has been responsible of the Laboratory Cognitive Systems, Institute for High Performance Computing and Networking (ICAR), National Research Council (CNR), Italy. He is currently a Researcher with the ICAR, CNR. He has been involved in different national and European projects. He has authored more than 100 peer-reviewed articles on international journals and conference proceedings. His current research interests include artificial intelligence (AI) and are focused on AI algorithms and techniques, mixing deep learning and knowledge-based technologies, for building intelligent systems able to converse, understand natural language and answer to questions, with emphasis on the distributional neural representation of words and sentences, and on specific natural language tasks such as part of speech tagging, sentence classification, and open information extraction. He is also a member of the editorial board of some international journals. He has been on the program committee of many international conferences and workshops.

...



GIUSEPPE DE PIETRO is currently the Director of the Institute for High Performance Computing and Networking, National Research Council (CNR), and also an Adjunct Professor with the College of Science and Technology, Temple University, Philadelphia, PA, USA. He has been actively involved in many European and national projects, with industrial co-operations. He has authored more than 200 scientific articles published in international journals and conferences.

Open Access funding provided by 'Consiglio Nazionale delle Ricerche-CARI-CARE' within the CRUI CARE Agreement

His current research interests include cognitive computing, clinical decision support systems, and software architectures for e-health. He is also a KES International Member. He is also involved in many program committees and journal editorial boards.

11043